Author's response to reviews

Title: Inhibitory effect of ginsenoside Rg3 combined with gemcitabine on angiogenesis and growth of lung cancer in mice

Authors:

Tai G Liu (liutaiguo031@126.com)
Ying Huang (huangying-68@qq.com)
Dan D Cui (huaximengqq@sina.com)
Xiao B Huang (chinahxb@hotmail.com)
Shu H Mao (17009391@qq.com)
Ling L Ji (89532810@qq.com)
Hai B Song (499805605@qq.com)
Cheng Yi (yicheng6834@163.com)

Version: 2 Date: 29 April 2009

Author's response to reviews:

28-Apr-2009

Dear Editor,

In the revised manuscript, we did some changes according to the comments and suggestions of referees and asked a friend majoring in English to improve the language (all modifications made are highlighted in red). In addition, we replied to the questions of the referees.

Sincerely yours,
Cheng Yi